Author's response to reviews

Title: Nursing Heroism in the 21st Century

Authors:

Philip Darbyshire (philip@philipdarbyshire.com.au)

Version: 3 Date: 28 January 2011

Author's response to reviews: see over
Final revision and responses to questions RE MS

MS: 1222912468483671
Nursing Heroism in the 21st Century
Philip Darbyshire

1. There were no reviewer’s comments to address.

2. I have substituted one word in the text: inserting ‘codes’ in place of ‘ethos’ on page 5, paragraph 3.

3. I have inserted a note acknowledging NET’s permission to cite a section from the In Pres editorial at Reference [30] and included the information that this is now available ‘online’.

4. I have checked the formatting and references an all seems correct.

5. I have uploaded the final revised version of the paper on to BMC.

Philip Darbyshire, 27th January 2011